Author’s response to reviews

Title: Implementation of an Older Person’s Nurse Practitioner in rural aged care in Victoria, Australia: a qualitative study

Authors:

Kaye Ervin (ervink@humehealth.org.au)

Carol Reid (reid.c@unimelb.edu.au)

Anna Moran (anna.moran@unimelb.edu.au)

Cynthia Opie (copie@erh.org.au)

Helen Haines (hmhaines61@gmail.com)

Version: 1 Date: 19 Jun 2019

Author’s response to reviews:

Victoria,

Australia is now included in the title of the paper

Referencing is now amended to be Vancouver style Lead author is now registered with capitals

The year (2018) of the study is now included in the abstract and text.